Author's response to reviews

Title: LEARN 2 MOVE 7-12 years: a randomized controlled trial on the effects of a physical activity stimulation program in children with cerebral palsy

Authors:

- Leontien Van Wely MSc (l.vanwely@vumc.nl)
- Jules G Becher MD, PhD (jg.becher@vumc.nl)
- Heleen A Reinders-Messelink PhD (H.A.Reinders@revalidatie-friesland.nl)
- Eline Lindeman MD, PhD (E.Lindeman@umcutrecht.nl)
- Olaf Verschuren PhD (o.verschuren@dehoogstraat.nl)
- Johannes Verheijden (johannes@bosk.nl)
- Annet J Dallmeijer PhD (A.Dallmeijer@vumc.nl)

Version: 2 Date: 14 October 2010

Author's response to reviews:

Please find enclosed a revised version of the study protocol of the LEARN 2 MOVE 7-12 study.

The following revisions have been made:
- all subheadings in the abstract have been removed according to the guidelines for authors. For that reason we changed the structure of the methods/design section of the abstract a bit, by starting with describing the design of the study.
- the output style of the reference list has been changed to the output style of BioMed Central